Discovering market segments for hunted wild game meat

Authors: Maria Elena Marescotti^a; Vincenzina Caputo^b; Eugenio Demartini^a, Anna Gaviglio^a;

^a Department of Health, Animal Science and Food Safety - Università degli Studi di Milano – Via Celoria, 10, 20133 Milano - Italy

^b Department of Agricultural, Food, and Resource Economics - Michigan State University

Authors' e-mail addresses:

Maria Elena Marescotti: maria.marescotti@unimi.it

Vincenzina Caputo: vcaputo@msu.edu

Eugenio Demartini: eugenio.demartini@unimi.it

Anna Gaviglio: anna.gaviglio@unimi.it

Corresponding author:

Maria Elena Marescotti - maria.marescotti@unimi.it tel. 0039-0250316488 - fax 0039-0250316486

Discovering market segments for hunted wild game meat

Abstract:

Recent years have seen a notable increase in the popularity of hunted wild game meat (HWGM) among consumers. This has led to a growing number of emerging markets for HWGM in many developed countries, including Europe. However, expansion of these markets is often hampered by the lack of a professional supply chain. The profitability of a supply chain would depend on consumer willingness to purchase HWGM products. This paper aims to (1) segment consumers based on their general attitudes towards HWGM, their perceptions of its safety, animal welfare, orientation concerning wildlife-related values, hunting activities, objective knowledge and socio-demographic factors and (2) assess whether these general attitudes affect consumer intentions to purchase HWGM products. To achieve our objective, a random sample of Italian consumers was recruited. Three different consumer segments were identified: pro-animal consumers, disoriented consumers, and hunted wild game meat eaters. Our findings highlighted an important lack of knowledge amongst consumers.

Keywords: Hunted wild game meat; hunting; cluster analysis; Italian consumer; consumer
 segmentation

1. Introduction

In recent years, there has been a notable increase in the popularity of hunted wild game meat¹ (HWGM) among consumers (Hoffman & Wilkund, 2006; Atanassova et al., 2008; FAO, 2007; Goguen et al., 2018). This has led to a growing number of emerging markets for HWGM in many developed countries, including Europe. To illustrate this, although there are no data available concerning the HWGM sector in Europe, the last European Commission study on minor meats (EC, 2014) reported that, in some member states, there is a well-established market for this type of product. For example, in France, about half of the game meat processors promote a "Game from French Hunters" brand (Gibier de Chasse - Chasseurs de France²). Similarly, in Spain, the Interprofessional Association for Hunted Game "Asiccaza" (Asociación Interprofesional de la Carne de Caza³) promotes wild game products that mainly come from hunting. However, the expansion of HWGM markets is often hampered by the lack of a structured food supply chain. This is especially the case in Italy where, despite flourishing populations of wild animals (Ramanzin et al., 2010), there is still a limited market for HWGM products (Gaviglio et al., 2017; Giacomelli & Gibbert, 2018).

Nonetheless, the creation of an Italian food supply chain for HWGM would generate market incentives that are expected to improve hunting practices and the management of wildlife overpopulation at no cost to the public. In addition, it would also generate new sources of income for populations living in marginal and rural mountain areas (Gaviglio et al., 2018). However, whether a supply chain for HWGM is economically sustainable depends on whether there is a demand for these products, which in turn depends on how consumers perceive them. In this regard, if obtained under strict and regulated hunting practices, HWGM embeds a number of quality features that may appeal to modern consumers when purchasing meat products (Hoffman & Wiklund, 2006; Ljung et al., 2012;

¹ In this paper the term "hunted wild game meat" refers to meat from large wild ungulates (e.g. wild boar – Sus scrofa,

red deer – *Cervus elaphus*, roe deer – *Capreolus capreolus*, chamois – *Rupicapra rupicapra*, mouflon – *Ovis orientalis*) obtained through hunting activities.

^{115 &}lt;sup>2</sup> http://chasseurdefrance.com/charte-gibier-de-chasse-chasseurs-de-france/

³ http://www.asiccaza.org/

Hoffman, 2013; AC Nielsen, 2016). For instance, HWGM may be considered organic and grass-fed meat because wild animals are, by definition, free roaming (Hoffman & Wiklund, 2006). In relation to free roaming, HWGM production may be considered an environmentally friendly livestock system (Thogmartin, 2006; Wiklund et al., 2014), representing a sustainable alternative to intensive livestock production for beef, pork or poultry (Ahl et al., 2002; Bureš et al., 2014; Thulin et al., 2015). 130 44 Additionally, as wild animals follow their natural grazing behaviour, hunting guarantees the 132 45 maximum level of animal welfare. Moreover, HWGM retains high nutritional values, with a low fat and cholesterol content, favourable n3:n6 fatty acid ratios and a high mineral content (Higgs, 2000; Rule, 2002; Webb, 2003; Hoffman & Wiklund, 2006; Bureš et al., 2014). Finally, if consumed in the area within which it has been hunted, HWGM embeds quality features associated with local food products (e.g., produced and consumed "from forest to fork") (Cerulli, 2012). However, despite these benefits, consumer valuation of HWGM can be hampered by other factors, such as environmental and ethical concerns regarding hunting practices (Ljung et al., 2012; Byrd et al., 2017). For instance, 147 52 certain consumers with pro-environmental and pro-animal attitudes may perceive hunting practices 149 53 as risky for the maintenance of wild animal species, or as a cruel activity committed against vulnerable living creatures (Kellert & Berry, 1987). Hence, consumers' perceptions of HWGM products may also be affected by concerns arising from individual attitudes towards hunting, animal welfare, and wildlife value.

To date, several studies have focused on the determinants for the consumption of meat from different animal species (Verbeke & Viaene, 1999; Grunert et al., 2004; Verbeke & Vackier, 2004; Angulo & Gill, 2007; Bonne et al., 2007; Pieniak et al., 2008; Verbeke et al., 2010; Pieniak et al., 164 60 2010b; Van Loo et al., 2010; Sepúlveda et al., 2011; Font-i-Furnols & Guerrero, 2014; Lusk & **61** Tonsor, 2016, among others). However, the economics and marketing literature on HWGM is still ¹⁷⁰ 63 limited. This study aims to (1) segment consumers based on their general attitudes towards HWGM, their perceptions of its safety, animal welfare, orientation concerning wildlife-related values, hunting

activities, objective knowledge and socio-demographic factors and (2) assess whether these general attitudes affect consumer consumption of HWGM products.

Our study addresses the need to broaden knowledge of consumers' attitudes towards HWGM and explores whether consumers would support a professional supply chain for HWGM, offering more information to stakeholders (i.e., hunters, processors, and retailers) to develop products and marketing strategies that effectively target individual consumer needs. Findings from this study can help policy makers to design new strategic interventions for the management of wild ungulate populations and the organization of professional supply chains for local HWGM products.

The remainder of the text is organized into four sections. Section 2 presents a literature review focused on consumers' attitudes towards HWGM. Section 3 presents the method and procedures adopted, including the survey framework (3.1), data collection and survey instrument (3.2) and statistical analysis (3.3.). Section 4 provides and discusses the results, while section 5 provides a summary of the research and some conclusions.

2. Consumers' attitudes towards hunted wild game meat: background

Most of the existing literature devoted to wild game meat consumption is primarily descriptive and focuses on non-European countries, such as Africa and Australia. With regard to the African context, a number of studies have focused on African consumers' perceptions and purchase behaviour for products derived from local species, such as springbok (Antidorcas marsupialis), blesbok (Damaliscus pygargus phillipsi), kudu (Tragelaphus strepsiceros), zebra (Equus burchelli), blue wildebeest (Connochaetes taurinus), impala (Aepyceros melampus) and gemsbok (Oryx gazelle) (Hoffman et al., 2003; Hoffman et al., 2004; Hoffman et al., 2005; Hoffman & Wiklund, 2006; Swanepoel et al., 2016). The results from these studies generally indicate that, despite the potential of wild game meat, African consumers and tourists are ill-informed regarding the positive attributes of game meat. This may be because producers and marketers are not doing enough to promote this 239 meat. In Australia, Hutchinson et al. (2010) performed a sensory analysis to investigate consumers' 90 240 241 evaluation of farmed red deer (Cervus elaphus) and fallow deer (Dama dama). The results from this 91 242 243 study suggest that Australian consumers prefer red deer meat compared to fallow deer meat and that 92 244 245 their perception of venison quality is mainly influenced by the method of carcass suspension, which 93 246 247 can enhance the meat's tenderness and succulence. However, due to the characteristics of HWGM 94 248 249 products, the results from these international studies cannot be extended to other geographical 250 95 251 252 96 contexts, such as Europe (Tomasevic et al., 2018), for a number of reasons. First, game meat 253 254 consumption is strictly related to the local available species and to their population size. Unlike Africa 97 255 256 98 and Australia, in Europe the most representative large wild ungulate species are wild boar (Sus scrofa) 257 258 99 and red deer (Cervus elaphus) (Hoffman & Wiklund, 2006; Hofbauer et al., 2010; Tomasevic et al., 259 260 2018). Second, some cultural differences across countries must also be considered. The acceptance 100 261 262 of hunting practices, in fact, is strictly connected to socio-cultural heritage, and depending on this ₂₆₃101 264 factor, harvesting and culling wild animals may or may not be ethically accepted (Mayfield et al., 265 102 266 2007; Willebrand, 2009; Ljung et al., 2012; Byrd et al., 2017; Gamborg & Jensen, 2017; Goguen et 267 103 268 269 104 al., 2018). 270

²⁷¹ 105 Therefore the question arises, what do we know about European consumers' perceptions of 272 273 106 HWGM? To date, there are only five studies assessing European consumer attitudes and purchase 274 275 276 **107** behaviour towards HWGM products. Tomasevic et al. (2018) have recently published the most 277 278 **108** exhaustive study on European consumers of hunted game meat. By using a cluster analysis, the 279 authors investigated consumers' perceptions, attitudes and perceived quality of game meat in ten ₂₈₀ 109 281 European countries (e.g., Czech Republic, Poland, Slovakia, Croatia, Albania, Bosnia and 282110 283 Herzegovina, Bulgaria, Former Yugoslav Republic of Macedonia, Montenegro and Serbia). The 284111 285 286 112 results from this study indicate that the consumption rate for game meat in the ten European countries 287 ²⁸⁸113 is influenced by a number of factors, such as location, age, and gender. For example, the authors 289 ²⁹⁰114 found that the consumption of HWGM is higher in South East European countries, and more popular 291 292 293 115 among men and older consumers.

294 295

²⁹⁸ 299 **116** Within the remaining literature, studies have just focused on consumers from one country. 300 117 Ljung et al. (2012) investigated the association between frequency of game meat consumption and 301 302 ₃₀₃118 attitudes towards hunting through an online survey sent to a random sample of Swedish residents. 304 Their findings suggest that game meat consumption and social relationships (i.e., having friends or 305 119 306 parents who hunt) were the key factors associated with positive attitudes towards hunting. Bodnar et 307 120 308 al. (2010) researched Hungarian consumers' demand for meat from game animals (red deer, fallow 309 121 310 311 122 deer, roe deer, wild boar, hare, pheasant and mallard duck) and found differences between a sample 312 ³¹³123 of respondents living in cities and those living in rural areas. Moreover, authors report that people 314 ³¹⁵ 316 124 with negative attitudes towards game meat are vegetarian or refuse its consumption for emotional 317 318 **125** reasons. With reference to the Polish context, Kwiecinska et al. (2017), using a model based on a 319 ₃₂₀126 logistic regression for predicting the consumption of wild game, demonstrated that purchase intention 321 increases with greater availability and the provision of higher quality. The results also showed a ₃₂₂ 127 323 higher propensity to change eating habits towards the consumption of game in men, city dwellers and 324 128 325 those who evaluated their own knowledge on nutrition and diet higher than others. Finally, the most 326 129 327 328 130 recent literature on European consumers' attitudes and purchase behavior for HWGM analyzes 329 ³³⁰131 consumer preferences for different types of preparation of red deer meat and beef in a northern Italian 331 ³³²132 population sample (Demartini et al., 2018). The authors conclude that considering consumers' 333 ³³⁴ 335 **133** attitudes towards wild game meat and hunting in their model significantly improved the interpretation 336 ₃₃₇134 of their results and allowed them to identify a niche market for red deer carpaccio (i.e. sliced fresh 338 raw meat). Although Demartini et al. (2018) found interesting results in relation to consumer ₃₃₉ 135 340 attitudes, the authors only included two of the multiple attitudinal dimensions relating to meat 341 136 342 consumption and were restricted in generalizing their results to the entire Italian population due to a 343 137 344 limited, non-representative sample. Thus, collectively, these prior studies suggest that consumer 345 138 346 ³⁴⁷ 139 attitudes impact on consumption of HWGM, yet there is more to be learned from a comprehensive 348 ³⁴⁹140 study specifically designed to assess the relationship between an array of consumer attitudes and 350 351 352 141 HWGM purchase behavior.

353 354

3. Method and procedures

3.1 Survey framework

To segment consumers of HWGM, we identified the attitudes and perceived risks that may influence consumer consumption of this type of meat and divided these attitudinal constructs into two groups. The first attitude group directly relates to the HWGM product and explores (1) consumers' general attitudes towards the product and (2) perceptions of HWGM safety. On the other hand, the second group gathers consumers' attitudes towards the supply chain for HWGM, such as (3) animal welfare, (4) attitudes towards hunting and (5) consumer wildlife-related values. Furthermore, to describe the segments, socio-demographic characteristics and consumers' objective knowledge about HWGM are considered in the analysis. Figure 1 shows a schematic representation of the variables under study.

Figure 1: Schematic representation of the variables investigated



3.2 Data collection and survey instrument

Data were collected using an online survey (Appendix A) sent to a random sample of Italian consumers during February and March 2018. The data collection was carried out using the Qualtrics®

416 417 159 online survey platform. Qualtrics is a leading world provider of survey samples⁴. Consumers were 418 419⁻¹⁶⁰ excluded from the survey if they did not buy meat during the three months preceding the survey and 420 ₄₂₁ 161 if they were aged under 18. The total sample comprised 1,029 respondents. 422

The survey consisted of a questionnaire containing closed-ended questions organized into four 423 **162** 424 sections following the framework presented in Figure 1. The first section aimed at detecting 425 163 426 consumption habits and objective knowledge about HWGM. To detect consumption habits, 427 164 428 ⁴²⁹ 165 respondents were asked whether they had consumed wild game meat in the last year (yes/no). To 430 ⁴³¹ 166 measure objective knowledge, we developed a series of true/false questions based on the literature on 432 433 434 167 the various empirical applications mentioned above. The following three items, out of the five, were 435 436 **168** correct: "In Italy, populations of wild ungulates (red deer, wild boar, roe deer, chamois) are growing 437 ₄₃₈ 169 rapidly"; "Consuming game meat (red deer, wild boar, roe deer, chamois) is an ethical choice, 439 440 **170** sustainable and with a very low environmental impact"; "Currently in Italy, hunting is 'controlled', 441 in fact it is subject to well defined rules concerning culling, species, times and places". The remaining 442171 443 two were false, and they were as follows: "Hunters are not considered producers by Italian law; 444 172 445 446 173 therefore they cannot sell the meat they hunt" and "Game meat (red deer, wild boar, roe deer, and 447 ⁴⁴⁸174 chamois) has a lower protein content and higher fat and cholesterol content than beef". 449

450 175 The second section of the questionnaire included questions aimed at capturing the consumers' 452 453 **176** general attitudes towards HWGM products and their perceptions of HWGM safety. Consumers' 455 **177** general attitudes towards game meat (ATT) were assessed by asking respondents to describe their ₄₅₇ 178 overall feelings when thinking about the consumption of HWGM products. More specifically, we adapted the version of the "general attitudes scale" proposed by Olsen et al. (2007) (scored on five-459179 point semantic differential scales), which has been widely used in the literature to assess consumers' 461 180

471 472

451

454

456

458

460

462 463 464

⁴⁶⁵ ⁴ Qualtrics is a world leading provider of survey samples. The sampling procedures are certified for the transparency of 466 the online process by the Qualtrics Esomar28 and through a variety of quality systems certifications, such as ISO 20252 467 management systems standards, Media Ratings Council, among others. Although online surveys are increasingly used in 468 consumer food choice studies, they may be subject to selection bias issues (Canavari et al., 2005; Windle & Rolfe, 2011; 469 Guimarães et al., 2015; Ripoll et al., 2015) due to the exclusion of individuals who do not use the internet. This might 470 induce slight differences between the general population and the sampled population.

attitudes towards food in general (Olsen, 2001, Honkanen et al., 2006; Pieniak et al., 2010b; Pérez-Cueto et al., 2011; Van Loo et al., 2013), and meat in particular (Berndsen & Van der Pligt, 2004; Almli et al., 2013). To capture consumers' perceptions towards wild game meat safety (SAF), we used the five-point semantic differential scale proposed by Almli et al. (2013) and considered six bipolar adjectives.

The third section of the questionnaire sought to highlight the typical factors that are expected to influence attitudes towards meat, in general, and HWGM in particular. To illustrate, previous studies have shown that attitude towards animal welfare (AW) is an important factor affecting consumer purchase intention for meat in general (Wong & Aini, 2017; Van Wezemael et al., 2010). Accordingly, we included the eight-item scale proposed by Kendall et al. (2006), which takes into account the general ethical issues linked to animal production, as well as questions relating to the different uses of animals. For each item, a 5-point interval scale ranging from "strongly disagree" to "strongly agree" was used. Additionally, we captured consumers' attitudes towards hunting (AH) by using the Likert scale proposed by Ljung et al. (2012). This scale is based on 9 items concerning hunting activity and hunters' behaviour. For each item, respondents were asked to express their agreement using a scale ranging from 1= strongly disagree to 5= strongly agree. Finally, since HWGM is obtained from wild animals, consumers' attitudes to this type of meat can be influenced 197 by their attitudes towards wildlife. Accordingly, the 8-item Likert scale proposed by Hrubes et al. (2001) was used to measure consumer wildlife-related values (WV). For each item, a 5-point scale ranging from 1= *strongly disagree* to 5= *strongly agree* was used.

Finally, the fourth section of the questionnaire collected the socio-demographic characteristics

518201 519 of the sample, including gender, age, region of residence, education level and income. Table 1 520 **202** 521 522 203 provides an overview of the socio-demographic characteristics of the sampled population and the 523 ⁵²⁴204 actual Italian population (Italian National Institute of Statistics data, 2018). 525 ⁵²⁶ 205 527

5	O I		, ,
6		Total	Italian
		1 Otal complo	nonulation*
		sample	
		(%)	(%)
	Gender		
	Male	49.08	49.82
	Female	50.92	50.18
		••••	00110
	Aga guoup		
	Age group	7 59	7 0 1
	10-22 yrs	7.50	/.01
	23-33 yrs	25.27	22.75
	36-55 yrs	51.31	48.59
	56-65 yrs	15.84	20.84
	Geographical region of residence		
	Northeast Italy	26 34	19.05
	Northwest Italy	18 46	26.22
	Southern Italy and Islands	27.50	20.22
	Southern hary and Islands	57.30	54.40
	Central Italy	17.69	19.85
	Education Level completed		
	Elementary School	0.10	5.74
	Middle School	8.07	31.95
	High School	55.00	44.48
	University and Postgraduate	36.05	17.83
	Other	0.78	
		0.70	
	Avarage household income		
	Average nousenoia income	<i>5</i> 1.00	1
	Low	51.99	n/a
	Medium	43.63	n/a
	High	4.37	n/a
207 208 209	* Source: Italian National Institute of Statistics data (IST) aged between 18 and 65 years.	AT 2018). Percentages are calcula	ated for the population
210	Of the 1,029 respondents in the study, according	to the national population	on, females were sl
11	over-represented (50.9%). In the sample, 51.3%	were aged between 36	5 and 55; 44.8% o
212	interviewed sample live in the northern part of I	taly, while 37.5% of the	e respondents were
213	southern Italy (islands included), and 17.7% were f	rom the central part of Ita	ly. Concerning educ
14	level, the sample was slightly biased towards bette	er-educated participants,	which may be due

Table 1. Socio demographic characteristics of the sample (Sample size n=1,029)

 $^{586}_{587}$ use of the online survey method. In fact, 55.0% of the respondents had completed college and 36.0%

⁵⁹³ 594</sub>216 had obtained a degree or a master's degree. Finally, most interviewees (51.99%) had self-reported 595 596**217** low financial status. 597 ₅₉₈218 599 600219 3.3 Statistical Analysis 601 The data were analysed using IBM SPSS Statistics (SPSS Inc., Chicago, IL). Survey data were first 602220 603 subjected to a descriptive analysis to provide a synthetic description of the main characteristics of the 604221 605 606 222 sample interviewed. 607 ⁶⁰⁸223 In addition, following the previous literature concerning consumer attitude studies (Roininen 609 ⁶¹⁰224 611

et al., 1999; Verbeke & Vackier, 2004; Vanhonacker et al., 2007; Van Wezemael et al., 2010; ⁶¹² 225 Vanhonacker et al., 2013; De Graaf et al., 2016), we also explored the internal reliability of the five 615²²⁶ multi-item attitudinal scales used in our survey instrument (e.g. ATT, SAF, AW, AH, WV). We did so by using the Cronbach's α (Cronbach, 1951; Peterson, 1994) and considering 0.6 as the threshold 617**227** value for a satisfactory scale (Verbeke & Vackier, 2004). Further, we expanded our analysis on the 619228 621 229 relationship underlying the set of variables for the five multi-item attitudinal scales through the ⁶²³230 execution of a principal component analysis (PCA) with varimax rotation (Malhotra, 1999).

⁶²⁵231 Next, we performed a cluster analysis (CA) to identify groups of consumers with similar 626 627 628**232** attitudes towards HWGM. Two steps were followed. In the first step, we applied the Hierarchical 629 ₆₃₀233 Clustering (Verbeke & Vackier, 2004; Dimech et al., 2011; Aprile et al., 2015) and the TwoStep 631 Cluster procedure (Bacher et al., 2004) to determine the optimal number of clusters. In the second ₆₃₂234 633 step, we used the optimal number of clusters derived from the first step to perform a non-hierarchical 634235 635 k-means cluster analysis (CA). Moreover, bivariate analyses were performed to explore whether the 636236 637 638237 identified clusters differ in terms of socio-demographics, consumption habits and knowledge about 639 ⁶⁴⁰238 HWGM. These analyses were performed using the one-way ANOVA with Dunnett's T3 post hoc 641 ⁶⁴²239 comparison of means, and cross-tabulation with χ^2 and Kruskal-Wallis statistics. Finally, we adopted 643 644 ₆₄₅240 the approach suggested by Dimech et al. (2011) and estimated a probit model to explore whether

646 647

591 592

613 614

616

618

620

622

624

belonging to a particular consumer segment identified by the CA relates to different HWGM consumption patterns.

4. Results and Discussion

4.1 Descriptive analysis

4.1.1 Consumption of wild game meat

The results of the analysis revealed that the majority (61.1%) of the respondents had consumed wild game meat at least once in the last year (termed "wild game meat eaters"). In contrast, 38.9% of the interviewees claimed that they had not consumed it in the last year (termed "wild game meat noneaters").

4.1.2 Consumers' objective knowledge about HWGM

Descriptive statistics for the objective knowledge variables are shown in Table 2. The table reports the total score for objective knowledge, which was created by summing the number of correct answers to the five true/false questions. The average number of correct answers was 2.66 on a five-point scale, indicating that, on average, respondents answered about half of the questions correctly. Of the sample, only 5% of the respondents answered all five true/false questions correctly, whereas 30% and 27% answered three and two correctly, respectively. More specifically, the majority of the sample interviewed (74%) answered correctly to the false statement "Game meat (red deer, wild boar, roe deer, chamois) has a lower protein content and higher fat and cholesterol content than beef", whereas most of the respondents (73.3%) failed to provide a correct answer to the false statement "Hunters are not considered producers by the Italian law; therefore they cannot sell the meat they hunt". These results highlight the need to increase consumers' awareness about HWGM. Overall, about 53% of the all the implied answers to statements were correct.

- 704

Statements	Correct (%)
Statement 1	55.49
Statement 2	26.72
Statement 3	74.05
Statement 4	60.64
Statement 5	70.26
Total number of correct answers	53.18
Respondents' number of correct answers to 5 state	ements
0	2.92
1	14.29
2	27.11
3	30.42
4	20.21
5	5.05

₇₄₃267 4.1.3 Consumers' general attitudes and perceptions of HWGM safety

745²⁶⁸ The general attitudes construct (ATT) was calculated as the average score across the six items. Respondents showed a general positive tendency towards the consumption of HWGM products ₇₄₉270 (mean value= 3.18 on a 5-point scale; SD= 1.03). The results of the general attitudes scale are reported in Figure 2.



Figure 2. Consumers' general attitudes (ATT)

As shown in Figure 2, the semantic differential item unfavourable/favourable showed the highest percentage of strongly positive and strongly negative feelings, with 15.7% of respondents favourable and 13.7% unfavourable. The analysis of the results shows that the group of consumers with the lowest attitudes towards HWGM, who indicated a dislike of the product, consider it terrible and feel bad when thinking about it; and the consumer group with the highest attitudes, who like HWGM and consider it good and delightful, have a similar sample size.

The perception of the HWGM safety (SAF) construct was computed as the average across the ⁸⁰⁹282 six items in the scale. As seen in Figure 3, respondents generally showed slightly positive feelings ⁸¹¹ 283 towards HWGM safety (mean value= 3.13, SD= 0.91). Across all the semantic items on the scale, ⁸¹³,284 between 35.7% and 44.7% of the respondents identified as uncertain; this result can be expected 816 285 considering respondents' low level of objective knowledge. Similar to the general attitudes towards ₈₁₈286 HWGM, the analysis of safety perceptions revealed the presence of two groups of respondents who have strongly negative ("suspicious", "uncomfortable", "unconfident") and/or strongly positive feelings ("satisfied", "optimistic", "confident") towards HWGM.

₇₉₉277

₈₀₁ 278

⁷⁹² 793**27**4

⁷⁹⁴275

⁷⁹⁶ 797**276**

Considering the limited body of literature concerning HWGM, consumers' general attitudes and perceptions of HWGM safety are not strictly comparable with other previous studies on this topic. However, the sample mean values relating to consumers' general attitudes and perceptions of HWGM safety are similar to the mean values reported by Van Wezemael et al. (2010) and by Almli et al. (2013) in a study concerning beef consumers in different European countries (respectively, Germany, Spain, France, and the UK in the first study, and Belgium and Norway in the second).

									
Pessimistic	10.01	9.43	40.	33		28.67		11.56	Optimistic
Unconfident	10.11	12.44	35	.67		31.29		10.50	Confident
Unsatisfied	8.94	14.77		40.72		23.71		11.86	Satisfied
Uncomfortable	10.59	13.02	2	44.61		23.	.42	8.36	Comfortable
Suspicious	10.88	14.6	7	38.78		26.3	4	9.33	Trustful
Worrying	9.82	12.93		44.70		23.:	52	9.04	At ease
0.	00	0.2	20 0.4	40 0.	60	0.3	80	1.0	00

Figure 3. Consumers' perceptions of HWGM safety (SAF)

4.1.4 Consumers' Attitudes towards animal welfare, hunting activity and wildlife-related values The third section of the questionnaire relates to factors that show a reasonable direct correlation with HWGM consumption, such as attitudes towards animal welfare, attitudes towards hunting and the perceived importance of wildlife.

Concerning attitudes towards animal welfare (AW), the construct was calculated as the average of the answers to the eight statements. Table 3 shows an overview of the results of the AW Likert scale.

886 887

⁸⁸⁸ 889</sub>305

Table 3. Consumers' attitudes towards animal welfare (AW)

Items	Strongly disagree (%)	Disagree (%)	Neither disagree nor agree (%)	Agree (%)	Strongly agree (%)	Mean
It is important that the food I normally eat has been produced in a way that animals have not experienced pain.	1.94	4.66	25.85	39.84	27.70	3.87
It is important that the food I normally eat has been produced in a way that animals' rights have been respected.	0.87	4.57	21.48	42.47	30.61	3.97
In general, humans have too little respect for the quality of life of animals.	1.55	5.05	23.62	40.33	29.45	3.91
Increased regulation of the treatment of animals in farming is needed.	1.75	3.50	22.25	39.26	33.24	3.99
Animal agriculture raises serious ethical questions about the treatment of animals.	2.14	4.66	25.17	41.01	27.02	3.86
As long as animals suffer pain, humans should not be able to use them for any purpose (R).	12.34	18.56	37.51	24.78	6.80	3.05
It is acceptable to use animals to test consumer products such as soaps, cosmetics and household cleaners (R).	35.47	22.06	25.56	14.19	2.72	3.73
Hunting animals for sport is an acceptable form of recreation (R).	42.08	18.85	23.13	13.41	2.53	3.85
Cronbach's $\alpha = 0.696$						

922 923 924 **307**

925

Note: R indicates items that have been reversely scaled in the analysis of the results; SD= Standard Deviation

⁹²⁶308 Sample results were quite high (mean value= 3.73, SD= 0.70), indicating that respondents have strong 927 ⁹²⁸309 attitudes towards animal welfare. Of the sample, 73.08% declared that it is important that the food 929 930 310 they normally eat has been produced in a way that respects animals' rights (mean value= 3.97), while 931 932 ₉₃₃311 72.5% agreed that increased regulation of the treatment of animals in farming is needed (mean value= 934 3.99). This finding is consistent with several consumer studies reporting consumers' concerns about ₉₃₅312 936 animal welfare (Frewer et al., 2005; Mayfield et al., 2007, Vanhonacker et al., 2007; Vecchio & 937313 938 Annunziata, 2012; Cembalo et al., 2016). Moreover, a substantial proportion of respondents (42.1%, 939314 940 mean value= 3.85) believe that hunting animals for sport is unacceptable, while only 15.9% of them 941 315 942

believe that recreational hunting is acceptable. These results are in line with previous studies on Italian consumers in relation to animal welfare issues (Harper & Henson, 2001; Mayfield *et al.*, 2007;

Vecchio & Annunziata, 2012).

Table 4 reports the results for attitudes towards hunting using a Likert scale (AH).

321	Table 4: Consumers'	attitudes	towards	hunting	(AH)

Items	Strongly disagree (%)	Disagree (%)	Neither disagree nor agree (%)	Agree (%)	Strongly agree (%)	Mean	SD
Hunting helps keep nature in balance.	15.06	23.52	39.94	18.85	2.62	2.70	1.02
Most hunters are well-prepared when they go hunting.	16.03	26.63	32.26	21.87	3.21	2.70	1.08
I see little wrong with harvesting animals for their meat, as long as the animal is not endangered.	12.73	14.48	31.58	33.92	7.29	3.09	1.13
Hunters are properly trained and follow hunting regulations.	19.53	25.75	36.83	15.26	2.62	2.56	1.05
Hunting is an important rural tradition.	14.19	16.72	36.73	27.31	5.05	2.92	1.10
I regard any kind of sport and recreational hunting as cruel to animals. (R)	4.37	8.16	25.66	30.81	31.00	2.24	1.11
Hunters often ignore safety rules. (R)	2.92	6.61	26.63	38.39	25.46	2.23	1.00
Hunters often harm animals, which then die a slow and painful death. (R)	2.53	5.25	32.94	33.82	25.46	2.26	0.98
I do not like people who hunt. (R) Crowbach's $a = 0.866$	6.03	10.98	38.87	22.16	21.96	2.57	1.13

Note: R indicates items that have been reversely scaled in the analysis of the results; SD= Standard Deviation

Attitudes towards hunting were notably quite low, with a mean value of 2.56 (SD= 0.74). Overall, ₉₈₈324 61.8% of respondents regarded sport or recreational hunting as cruel to animals (mean value= 2.24), 990 325 corroborating the results of the AW scale item regarding hunting animals for sport. On the other hand, 41.2% of them (mean value= 3.09) agreed with the acceptability of hunting game animals for food, ⁹⁹⁶328 as long they are not endangered, revealing that hunting is more acceptable to the public if it has some ⁹⁹⁸329 kind of utility (Gamborg & Jensen, 2017). Respondents generally expressed negative feelings and 330 concern towards hunters, stating that hunters often ignore safety rules (63.8%; mean value= 2.23),

1005	
$1006 \\ 331 \\ 1007$	often harm animals (59.3%; mean value= 2.26), are not properly trained and do not follow hunting
1008 1009 ³³²	regulations (45.3%; mean value= 2.56). Finally, approximately 44% of the respondents report
1010 101 333	disliking people who hunt (mean value= 2.57). Our findings are consistent with the findings of a
1012 101 334	previous study on Italian consumers' behaviour (Mayfield et al., 2007) and differ from the results for
1014	
101 335 1016	other European countries, such as Sweden (Mayfield et al., 2007; Ljung et al., 2012), where
101 7336 1018	consumers were shown to be strongly in favour of hunting. This result may be because, in Sweden,
101 9337 1020	hunting is not only a sport but also essential for food acquisition purposes (Mayfield et al., 2007),
102 338 1022	highlighting that consumers' attitudes towards hunting are strongly influenced by the socio-cultural
1023 3 39 1024	context and, in particular, the final purpose of the hunting activities (hunting for sport, hunting for
1025 1026 340	meat, hunting for wildlife population control, hunting for a trophy).
1027 1028 <mark>341</mark>	Table 5 shows consumers' wildlife-related value orientations (WV).
1029	
103042	
1031	
1032	
1033	
1034	
1035	
1030	
1037	
1030	
1033	
1041	
1042	
1043	
1044	
1045	
1046	
1047	
1048	
1049	
1050	
1051	
1052	
1053	
1054	
1055	
1056	
1057	
1058	
1059	
1061	10
1062	10
1002	

Table 5:	Consumers'	attitudes	towards	wildlife

			Neither			
Items	Strongly disagree (%)	Disagree (%)	disagree nor agree (%)	Agree (%)	Strongly agree (%)	Mean
It is important to manage populations of wildlife for the benefit of humans. (R)	5.73	12.24	34.01	35.37	12.63	2.63
I enjoy watching wildlife when I take a trip.	1.17	4.18	19.53	47.81	27.31	3.96
It is important to protect wildlife for future generations.	0.68	2.04	14.38	42.76	40.14	4.20
Hunting and fishing are cruel and inhumane to the animals.	7.39	16.13	42.27	22.06	12.15	3.15
I notice birds and wildlife around me every day.	3.89	11.47	31.97	38.19	14.48	3.48
People should not cause pain and suffering to wildlife, regardless of how much we may benefit.	3.21	7.58	27.31	31.88	30.03	3.78
It is important that we learn all we can about wildlife.	0.68	3.98	23.91	43.93	27.50	3.94
Animals should have rights similar to the rights of humans.	0.25	12.54	36.93	28.86	16.42	3.39
Cronbach's $\alpha = 0.830$						

109\$345 Note: R indicates items that have been reversely scaled in the analysis of the results; SD= Standard Deviation

Reported wildlife-related value orientations, computed as the average score across the items, were generally positive (mean value= 3.16; SD= 0.60), indicating that wildlife enjoyment and animal rights/management are important for the interviewees (Fulton et al., 1996). The majority of the respondents stated that they enjoy watching wildlife (75.1%; mean value= 3.96) and that it is important to learn all we can from wildlife (71.4%; mean value= 3.94). Moreover, almost 40.1% of the interviewees strongly agreed and 42.8% agreed that it is important to protect wildlife for future ¹¹¹353 1112 generations (mean value= 4.20). On the other hand, approximately half of the sample stated that it is 354 1114 important to manage populations of wildlife for the benefit of humans. From these results, a relevant 1116³⁵⁵ proportion of respondents have a "mutualist value orientation", believing that humans and wildlife are meant to coexist or live in harmony and, thus, that wild animals deserve rights similar to the rights 1118³⁵⁶

343

1068 **344**

of humans (Charles M. Russell National Wildlife Refuge & UL Bend National Wildlife Refuge, 2010; Gamborg & Jensen, 2016). In contrast, only a limited part of the sample had a "utilitarian value orientation" and thus believed that wild animals should be used for the benefit of humans and that hunting is acceptable (Gamborg & Jensen, 2016).

4.2 Reliability analysis and principal component analysis

The internal reliability consistency of the AW, ATT, SAF, AH, and WV scales was analyzed using the Cronbach's α test. The results were all higher than the 0.6 threshold value for a satisfactory scale, indicating that all the adopted scales are valid instruments to measure the proposed constructs. The lowest values were obtained for the animal welfare scale (AW α = 0.696), while the other scales had Cronbach's α -values higher than 0.8 (ATT α = 0.946; SAF α = 0.919; AH α = 0.866; WV α = 0.830).

To further investigate the structure and relevance of these scales in explaining consumer attitudes towards HWGM, we also performed a principal component analysis (PCA) using the varimax rotation (see Appendix B, Table B1). The PCA revealed eight components, whose internal reliability values (measured using the Cronbach's α test) are lower than the values calculated on the original attitudinal scales. This suggests that the constructs from the PCA are less consistent than the original scales, and indicates that techniques applied to factor scores (as, for example, CA) may perform worse than on original variables (Fiedler & McDonald, 1993). Moreover, removing the items that loaded on different factors, as suggested by De Graaf et al. (2016) and Verbeke & Vackier (2004), has not improved the internal reliability values of the constructs (see Appendix B, Table B2).

4.3 K-means cluster analysis

The cluster analysis was performed using the k-means method and the index of the items/constructs from the five original scales (e.g., ATT scale, SAF scale, AW scale, AH scale, WV scale). Prior to

applying the k-means method, we determined the optimal number of classes using the Hierarchical Clustering and Two-Step Cluster procedures. A total of three optimal clusters were identified⁵.

To verify the existence of significant differences between clusters, one-way analysis of variance (ANOVA) was performed to compare factor means. Considering their respective reported factor mean scores, clusters have been classified as pro-animal, disoriented, and HWGM eaters. The results of the cluster and of the one-way ANOVA analyses are summarized in Table 6.

Table 6. Final cluster average scores

	Cluster 1	Cluster 2	Cluster 3	F statistics
	(n= 168; 16.33%)	(n= 574; 55.78%)	(n=287;27.89%)	(F-test)
General attitudes	1.532ª	3.116 ^b	4.257°	1348.02 ***
Perception of HWGM safety	1.793ª	3.075 ^b	4.035°	855.54 ***
Animal welfare	4.253ª	3.753 ^b	3.382°	98.75 ***
Attitudes towards hunting	1.861ª	2.489 ^b	3.107°	223.81 ***
Wildlife related value	3.605 ^a	3.197 ^b	2.842°	105.64 ***
Cluster classification	Pro-animals	Disoriented	HWGM eaters	

Significance Levels ***p< 0.001; **p< 0.010; *p< 0.050

^{a,b,c} indicate significantly different means using one-way ANOVA and post hoc Dunnet T3 multiple comparison test (equal variances not assumed)

The ANOVA proved that all segments significantly differ for all the factors under study (p < 0.001). The highest differences among consumer groups were related to the factor "general attitudes". Figure 4 reports a graphical representation of the profiles for each cluster, while Figure 5 graphically reports

the results for each construct.

⁵ The results of the Hierarchical Clustering and Two-Step Cluster procedures are available upon request.



Figure 4. Final cluster profiles





¹³⁴401 To better describe the characteristics of the individuals in each class, we also designed post-¹³⁴³ 402 clustering profiles in relation to the most meaningful variables used in the survey, such as 1346 1346 demographic characteristics, consumption habits, and consumer knowledge.

To profile each of the segments identified in terms of socio-demographic characteristics, we performed a cross-tabulation analysis. Table 7 summarizes the socio-demographic profile of the three **6**105 identified clusters, as well as the statistically significant differences found between groups. **2406** Differences across segments were measured using the chi-square test or Kruskal-Wallis test, depending on the characteristics of the variables.

	Cluster 1	Cluster 2	Cluster 3			
Item	Pro-animal	Disoriented	HWGM eaters	p-v	alue	
	(%; n= 168)	(%; n= 574)	(%; n= 287)	(x2-test)	(Kru Walli	ıskal- is test
Gender						
Male	28.57	47.74	63.76	0.000 ***		
Female	71.43	52.26	36.24			
Age group						
18-22 yrs	8.33	9.41	3.48		0.106	
23-35 yrs	22.02	26.66	24.39			
36-55 yrs	55.36	47.74	56.10			
56-65 yrs	14.29	16.20	16.03			
Geographical region of res	idence					
Northeast Italy	14.88	17.77	21.95	0.090		
Northwest Italy	21.43	26.83	28.22			
Southern Italy and Islands	45.83	37.98	31.71			
Central Italy	17.86	17.42	18.12			
Education Level completed						
Elementary School	0.00	0.17	0.00		0.029	*
Middle School	10.71	8.01	6.62			
High School	57.14	56.62	50.52			
University and Postgraduate	31.55	34.15	42.51			
Other	0.60	1.05	0.35			
Average household income						
Low	62.50	54.18	41.46		0.000	**
Medium	33.93	41.99	52.61			
High	3.57	3.83	5.92			

Table 7. Socio-demographic profiles of the three identified clusters

Significance Levels: ***p< 0.001; **p< 0.010; *p< 0.050

Chi-square and Kruskal-Wallis statistics results showed statistically significant differences between the three consumer groups in terms of gender (p= 0.000), education level completed (p= 0.029) and average household income (p= 0.000).

The first cluster included 16.3% of the respondents (n= 168). These consumers were particularly concerned about animal welfare and gave a high value to wildlife (Table 7 and Figure 4). Moreover, they had very low attitudes towards HWGM, as well as towards hunting, and revealed strongly negative feelings regarding HWGM safety. Thus, this group was classified as *pro-animal consumers*. The results reported in Table 8 revealed that the members of this group were more likely to be female (71.4% of the total sample), with a middle school or high school diploma (67.85%), and with a low income (62.5%). Furthermore, there was a tendency (p= 0.090) for the respondents in this cluster to come from southern Italy and the Islands (45.8%).

The second cluster, classified as *disoriented consumers*, was the largest, accounting for 55.8% of the sample (n= 574). Respondents in this group, even if they revealed concerns about animal welfare and placed importance on wildlife enjoyment and animal rights, had positive attitudes towards HWGM and its safety. On the other hand, they had relatively low attitudes towards hunting (Table 7 and Figure 4). From a socio-demographic viewpoint, as shown in Table 6, consumers belonging to this cluster had a better financial status (54.2% had low and 42.0% medium financial status), and they mainly came from southern Italy and the Islands (38.0%). In contrast to the first cluster, this group contained better-educated participants (34.15% of them had obtained a degree or a master's degree). With a proportion of 47.7% males and 52.3% females, this cluster was the most balanced of the three. Finally, the third cluster (27.9% of the sample, n= 287) contained consumers with strongly positive attitudes towards the consumption of HWGM products and strongly positive feelings concerning its safety. Thus, this cluster was defined as *HWGM eaters*. These consumers reported the highest attitudes towards hunting and the lowest score related to animal welfare issues and wildlife value orientation. In contrast with the other groups, HWGM eaters were mainly males

436 1462 consumers with the highest education level; in fact, 42.5% of them reported having a university or 1464**437** postgraduate degree. Moreover, although not statistically significantly different, consumers in this group seemed to be more likely to come from northern (50.2%) and central Italy (18.1%) and to be **⁴38** aged over 36 years (72.0%). **8**39 In relation to consumer objective knowledge concerning hunted wild game meat (Table 8), 147@440 statistically significant differences were found between the three consumer clusters (p=0.000). Pro-animal consumers showed the lowest degree of knowledge (mean value= 1.90 correct answers out of the five statements), followed by disoriented consumers (mean value= 2.59). HWGM eaters 444 1479 demonstrated the highest degree of knowledge about HWGM (mean value= 3.24). These findings 1481 suggest that a poor level of consumer knowledge regarding HWGM could act as a barrier to its consumption. Consumers, in fact, are often not aware that hunting can be practised ethically and that

it is subject to strict and well-defined regulations. Our results are consistent with previous studies (Pieniak et al., 2010a; Van Loo et al., 2013) reporting that higher consumer knowledge about certain food products has a positive influence on their consumption.

(63.7%) with a medium or high financial status (58.5%). This cluster had the highest percentage of

		Cluster 1	Cluster 2	Cluster 3	
Item	Total sample	Pro-animals	Disoriented	HWGM eaters	F statistics
	(n= 1.029)	(n= 168)	(n= 574)	(n= 287)	(F-test)
Objective					
Knowledge					
(no. of correct					
answers to 5 statements)	2.66	1.90 a	2.59 b	3.24 °	0.000 **
a-b-c Indicate significan	ntly different means usin	ng Scheffé post hoc test (equal v	variances assumed)		
Finally	, we followed	l Dimech et al., (2	2011) to explore w	whether consumers	belonging t

Table 8. Objective knowledge related to hunted wild game meat among different clusters

specific segment or cluster, and with certain socio-demographic characteristics, are more or less likely

¹⁴⁵⁹435

to consume HWGM. More specifically, we estimated a probit model using the self-reported
consumption of HWGM in the last year as a dependent variable, and the consumers' sociodemographic characteristics as covariates. The variables are defined in table 9, while the probit
estimates are reported in table 10.

1527
458Table 9. Probit model variables

Variable	Description
Dependent varia	ble
HWGM Cons	Equals 1 if the respondent has consumed HWGM in the last year; 0 otherwise
Categorical Cove	ariates (factors)
Cluster	
Cluster 1	Equals 1 if the respondent belongs to the <i>Pro-animal</i> cluster. 0 otherwise (baseline)
Cluster 2	Equals 1 if the respondent belongs to the <i>Disoriented</i> cluster 0 otherwise
Cluster 3	Equals 1 if the respondent belongs to the <i>HWGM eaters</i> cluster, 0 otherwise
Gender	
Male	Equals 1 if the respondent is a male 0 otherwise (baseline)
Female	Equals 1 if the respondent is a female 0 otherwise
Area	Equals 1 if the respondent is a remaine, o otherwise
NEItoly	Equals 1 if the respondent lives in Northeast Italy, 0 otherwise (baseline)
INEItaly NUVItaly	Equals 1 if the respondent lives in Northwest Italy, 0 otherwise (basenine)
IN W Italy	Equals 1 if the respondent lives in Northwest Italy, 0 otherwise
Sitaly	Equals 1 if the respondent lives in Southern Italy and Islands, 0 otherwise
Citaly	Equals 1 if the respondent lives in Central Italy, 0 otherwise
Covariates	
Age	Respondent age. Equals 1 if the respondent is aged between 18 and 22 yrs, 2 if between 23 35 yrs, 3 if 36 55 yrs, 4 if 56 65 yrs
	Respondent education level Equals 1 if completed middle school 2 high school 3
Edu	University/Postgraduate 4 other kind of school
Inc	Respondent average household income Equals 1 if low 2 if medium 3 if high

1576

1616 1617

1577461 **Table 10. Probit estimates results**

T	amin la a	Caaff	Std Emer	Signifi	cance	**
v	ariables	Coeff.	Sta. Error	Z	P> z	
Cluster 2 - Disorie	ented	0.846	0.120	7.059	0.000 **	*
Cluster 3 - HWGM	A eaters	1.506	0.142	10.604	0.000 ***	*
Female		0.041	0.088	0.469	0.639	
Northwest Italy		0.194	0.131	1.479	0.139	
Southern Italy and	Islands	-0.022	0.110	0.205	0.838	
Central Italy		0.136	0.133	1.025	0.305	
Age		-0.127	0.053	2.423	0.015 *	
Edu		0.246	0.072	3.414	0.001 ***	*
Income		0.374	0.080	4.687	0.000 ***	*
Constant		-1.539	0.355	4.336	0.000 ***	*
Log Likelihood		-306.135				
Pseudo R-squared		0.48				

Note: Significance Levels: ***p < 0.001; **p < 0.010; *p < 0.050; for the categorical covariates baselines refer to the 1595 Table 8 1596

1597 1598⁴⁶² As expected, probit estimates show that the proportion of wild game meat eaters increases across 1599 consumer segments (p < 0.001) from Cluster 1 (pro-animal) to Cluster 3 (HWGM eaters), and that 1600463 1601 individuals are more likely to consume HWGM if they are younger (p < 0.050). Moreover, the 160**2**464 1603 probability of HWGM consumption increases with the individual's education level and household 1604465 1605 160@466 income (p< 0.001). Gender and geographical region of residence seem to have no effect on HWGM 1607 1608467 consumption. Finally, the statistical significance of the coefficients for clusters 2 and 3 indicate 1609 ¹⁶¹⁰468 differences in consumption patterns for HWGM across the three clusters or segments. According to 1611 1612 469 the Wald Chi-Squared Test (reported in table 10) the null hypothesis of equality between classes can 1613 1615**470** be rejected, indicating that the probability of consuming HWGM varies across consumer segments.

Table 10. Wald Tests across clusters 1618⁴⁷¹

1619			
1620		Wald	
1621	Hypothesis	Chi-	<i>p</i> -value
1622		square	
1623	Ho: Cluster $1 = $ Cluster 2	49,834	0.000 ***
1624	Ho: Cluster $1 = $ Cluster 3	112,443	0.000 ***
1625	Ho: Cluster $2 = $ Cluster 3	38,883	0.000 ***
1626	Significance Levels: ***n< 0.001	$1 \cdot **n < 0.010 \cdot *n$	< 0.050
1627	Significance Levels. p< 0.001	r, p<0.010, p	< 0.050
1628472			
1629			
1630			
1631			
1632			
1633			
1634			

5. Conclusions

This study profiles Italian consumers according to their attitudes and perceptions towards HWGM, socio-demographic characteristics and objective knowledge, and it assesses whether these factors affect consumption of HWGM products. Through a cluster analysis, three different consumer groups were identified and classified as pro-animal consumers, disoriented consumers and HWGM eaters. The three groups (clusters) showed significant differences with respect to their sociodemographic characteristics (gender, education level and average household income), consumption of HWGM and their level of objective knowledge. Cluster one, termed pro-animal consumers, was the smallest (16.3%) and included significantly more females, those who are less educated, and those with a low income who are particularly concerned about animal welfare and give high value to wildlife. This cluster showed very low attitudes towards HWGM, hunting, and HWGM safety, as well as the lowest degree of knowledge about HWGM. This group seemed the least interesting for future marketing strategies. The second and largest cluster (55.8%) was the intermediate group of disoriented consumers that mainly comprised consumers who were more educated, with a good financial status, who revealed concern about animal welfare and gave importance to wildlife enjoyment and animal rights, but had positive attitudes towards HWGM and its safety. On the other hand, these consumers had relatively low attitudes towards hunting. Moreover, the majority of the disoriented consumers (60.1%) reported having eaten HWGM in the last year, but showed a low degree of knowledge about HWGM. Considering all these factors, this consumer group is the most interesting for future targeted marketing strategies aimed at increasing HWGM consumption. Finally, the cluster of HWGM eaters (27.9% of the sample) contained mainly male consumers, consumers with the highest education level, and those with a medium or high financial status, with strong positive attitudes towards the consumption of HWGM products and strong positive feelings concerning its safety. These consumers reported the highest attitudes towards hunting and the lowest scores related to animal welfare issues and wildlife value orientation. Moreover, the majority of them (83.0)

reported having eaten wild game meat in the last year, and they had the highest degree of knowledge 1698 about HWGM.

1700**501** Our findings provide new insight for the development of a new market for Italian HWGM, highlighting the existence of an important lack of consumer knowledge that could act as a barrier to consumption. Several studies, in fact, have reported that higher consumer knowledge about certain food products has a positive influence on purchase and consumption (Brucks, 1985; Pieniak et al., 2010a; Van Loo et al., 2013; Prestamburgo & Sgroi, 2018). There is a need to better inform and educate citizens about the hunted wild game meat sector, as well as about HWGM. Consumers need to be informed about European hunting regulations, the role of hunting activities in the management of large wild animal overpopulation, and the impact of the high densities of these animals on ecosystems and human activities. On the other hand, consumers need to be aware of all the positive intrinsic features of HWGM. Only by improving consumers' knowledge about HWGM is it possible to increase the likelihood of its consumption. Further research is needed to provide new insights into the existence of consumer segmentation towards HWGM in different geographical areas and cultural contexts. Future studies on this topic should focus on the analysis of consumer preferences and willingness to pay for Italian hunted wild game meat products by adopting the discrete choice experiment (DCE) method.

1766

¹⁷⁵⁴517 **Funding:** 1755

175618 This research work has been funded by University of Milan – Department of Health, Animal Science 1757 and Food Safety (VESPA) "Piano di Sostegno alla Ricerca 2017 - Linea 2 Azione A)"; and 175819 ¹⁷⁵⁹**520** 1760 Fondazione Cariplo "Bandi Ambiente 2016" - Progetto "Processi di Filiera Eco-Alimentare - La 176**521** gestione di prodotto sostenibile per lo sviluppo dei territori alpini". 1762 5**22** 1763

1764 1765²³ **Bibliography**

- AC Nielsen (2016). Weighing consumers' growing appetite for "Clean" meat labeling. Fresh Facts 1767524 historical data, 2011-2015. Insights news, October 2016. 176525
- Ahl, A., Nganwa, D., Wilson, S. (2002). Public health considerations in human consumption of wild 176926 game. Annals of the New York Academy of Sciences, 969, 48-50. 177527
- Almli, V.L., Van Wezemael, L., Verbeke, W. (2013). One technology does not fit all: Profiling 177528 1773229 consumers of tender and tenderised beef steaks. Meat Science, 93, 361-370.
- 1773330 Angulo, A.M.; Gil, J.M. (2007). Risk perception and consumer willingness to pay for certified beef 177**5**31 in Spain. Food Quality and Preference, 18: (8), 1106-1117.
- 1775532 Aprile, M.C., Caputo, V., Nayga, R.M. (2015). Consumers' preferences and attitudes toward local 1776333 food products. Journal of Food Products and Marketing, 22: (1), 1-26.
 - Atanassova, V., Apelt, J., Reich, F., Klein, G. (2008). Microbiological quality of freshly shot game in Germany. Meat Science, 78, 414-419.
 - Bacher, J., Wenzig, K. & Vogler, M. (2004). SPSS twostep cluster: A first evaluation. (Arbeits- und Diskussionpapiere. 2, 2) Erlange-Nurnberg, University of Friedrich-Alexander, Chair of Sociology (retrieved August 25, 2007).
 - Berndsen, M., Van der Pligt, J. (2004). Ambivalence towards meat. Appetite, 42, 71-78.
- 177534 1778335 1779535 1779536 1780536 1780537 178538 178538 178539 178539 178540 Byrd, E., Lee, J.G., Widmar, N.J.O. (2017). Perceptions of hunting and hunters by U.S. respondents. Animals (Basel), 7: (11), 83. 178541
- Bodnar, K., Benak, A., Bodnarne Skobrak, E. (2010). Analysis of consumers preferences and 178042 attitudes on Hungarian game meat market (preliminary report). Lucrări Stiințifice, 53: (1), 9-12. 178543
- 1785544 Bonne, K., Vermeir, I., Bergeaud-Blackler, Verbeke, W. (2007). Determinants of halal meat consumption in France. British Food Journal, 109: (5), 367-386. 1789545
- 179646 Brucks, M. (1985) The effects of product class knowledge on information search behavior. Journal 179547 of Consumer Research, 12: (1), 1-16.
- 1792348 Bureš, D., Bartoň, L., Kotrba, R., Hakl, J. (2014). Quality attributes and composition of meat from 179349 red deer (Cervus elaphus), fallow deer (Dama dama) and Aberdeen Angus and Holstein cattle ¹⁷⁹⁴550 ¹⁷⁹⁵551 (Bos taurus). Journal of the Science of Food and Agriculture, 95, 2299-2306.
 - Canavari, M., Nocella, G., Scarpa, R. (2005) Stated willingness-to-pay for organic fruit and pesticide ban: an evaluation using both web-based and face-to-face interviewing. Journal of Food Products Marketing, 11: (3), 107-134.
- 1796 552 1797 553 1798 1799 554 1806 555 Cembalo, L., Caracciolo, F., Lombardi, A., Del Giudice, T., Grunert, K.G., Cicia, G. (2016). Determinants of individual attitudes toward animal welfare-friendly food products. Journal of Agricultural and Environmental Ethics, 29: (2), 237-254. 180⁵⁵⁶
- Cerulli, T. (2012). The mindful carnivore: a vegetarian's hunt for sustenance. New York, NY: 180257 Pegasus Books. 180\$58
- Charles M. Russell National Wildlife Refuge, UL Bend National Wildlife Refuge (2010). 180**5**59 Comprehensive Conservation Plan: Environmental Impact Statement. U.S. Fish and Wildlife 180560 180**561** Service. Northwestern University – Montana.
- 180**562** Cronbach, L. J. (1951). Coefficient alpha and the internal structure of tests. Psychometrika, 16: (3), 180863 297-334.

1809 1810

- ¹⁸¹³564 Dimech, M., Caputo, V., Canavari, M. (2011). Attitudes of Maltese consumers towards quality in fruit and vegetables in relation to their food-related lifestyles. International Food and Agribusiness Management Review, 14: (4), 21-36.
- 1815 1815 1815 1815 1816 1817 1817 1818 68 1818 69 De Graaf, S., Vanhonacker, F., Van Loo, E.J., Bijttebier, L.L., Tuyttens, F.A.M, Verbeke, W. (2016). Market opportunities for animal-friendly milk in different consumer segments. Sustainability, 8: (12), 1302.
- 182570 Demartini, E., Vecchiato, D., Gaviglio, A., Tempesta, T., Viganò, R. (2018). Consumer preferences for red deer meat: A discrete choice analysis considering attitudes towards wild game meat and 182571 hunting. Meat Science, 146, 168-179 182**272**
- European Commission Directorate General for Agriculture and Rural Development (2014). Study 182**\$73** Report.- Minor Meats. September 2014. LEI Wageningen UR. 182**574**
- 182575 FAO (2007) FAOSTAT. http://faostat.fao.org//.
- 182976 Fiedler, J., & McDonald, J. J. (1993). Market figmentation: Clustering on factor scores versus 182377 individual variables. AMA Advanced Research Techniques Forum.
- ¹⁸²⁸ 578 1829 Font-i-Furnols, M., Guerrero, L. (2014). Consumer preference, behavior and perception about meat 183079 and meat products: An overview. Meat Science, 98, 361-371.
- Frewer, L.J., Kole, A., Van De Kroon, S.M.A., De Lauwere, C. (2005). Consumer attitudes towards 183**580** the development of animal-friendly husbandry systems. Journal of Agricultural and 1832581 183382 Environmental Ethics, 18, 345-367.
- 1834583 Fulton, D.C., Manfredo, M.J., Lipscomb, J. (1996). Wildlife value orientations: A conceptual and 183584 measurement approach. Human Dimensions of Wildlife, 1: (2), 24-47.
- 1836 1837 1837 1838 1838 Gamborg, C., Jensen, F.S. (2016). Wildlife value orientations among hunters, landowners and the general public: A Danish comparative quantitative study. *Human dimensions of wildlife*, 21: (4), 328-344. 183587
- Gamborg, C., Jensen, F.S. (2017). Attitudes towards recreational hunting: A quantitative survey of 184588 184589 the general public in Denmark. Journal of Outdoor Recreation and Tourism, 17, 20-28.
- 184390 Gaviglio, A., Demartini, E., Marescotti, M.E. (2017). Opportunities and limitation from an Italian ¹⁸⁴³591 alpine case study. Calitatea – Access la Succes, 18: (2), 215-222.
- 1844 1845 1845 1845 1846 Gaviglio, A., Marescotti, M.E., Demartini E. (2018). The local value chain of hunted red deer meat: A scenario analysis based on a northern Italian case study. Resources, 7: (2), 34.
- Giacomelli, S., Gibbert, M. (2018). "He likes playing the hero I let her have fun shooting". Gender 184**594** games in the Italian forest during the hunting season. Journal of Rural Studies, 62: 164-173. 184**59**5
- Goguen, A.D., Riley, S.J., Organ, J.F., Rudolph, B.A. (2018). Wild-harvested venison yields and 184996 185**G97** sharing by Michigan deer hunters. Human Dimension of Wildlife, 23: (3), 197-212.
- 185598 Grunert, K.G., Bredahl, L., Brunsø, K. (2004). Consumer perception of meat quality and implications 1852399 for product development in the meat sector – a review. *Meat Science*, 66, 259-272.
- 1853 00 1854 001 1855 001 1855 002 1856 003 1856 04 1856 04 Guimarães, M.H., Nunes, L.C., Madureira, L., Santos, J.L., Boski, T., Dentinho, T. (2015). Measuring birdwatchers preferences: a case of using online networks and mixed-mode surveys. Tourism Management, 46: 102-113.
- Harper, G., Henson, S. (2001). Consumer concerns about animal welfare and the impact on food choice. CT98-3678 EU FAIR Final Report, 38 pp. http://europa.eu.int/comm/food/animal/welfare/eu fair project en.pdf. 185905
- Higgs, J.D. (2000). The changing nature of red meat: 20 years of improving nutritional quality. Trends 186006 in Food Science & Technology, 11, 85-95. 186**607**
- Hofbauer, P., Smulders, F.J.M, Vodnansky, M., Paulsen, P. (2010). A note on meat quality traits of 186208 pheasants (Phasianus colchicus). European Journal of Wildlife Research, 56: (5), 809-813. 1863609
- Hoffman, L.C., Crafford, K., Muller, N., Schutte, D.W. (2003). Perceptions and consumption of game 186**⁄61**0 186**511** meat by a group of tourtists visiting South Africa. South African Journal of Wildlife Research, 33: 186612 (2), 125-130.
- 1867
- 1868
- 1869

- ¹⁸⁷²613 Hoffman, L.C., Muller, M., Schutte, D.W., Crafford, K. (2004). The retail of South African game meat: current trade and marketing trends. South African Journal of Wildlife Research, 34: (2),123-134.
- 187314 187415 187515 187516 187616 187617 187618 Hoffman, L.C., Muller, M., Schutte, D.W., Calitz, F.J., Crafford, K. (2005). Consumer expectations, perceptions and purchasing of South African game meat. South African Journal of Wildlife Research, 35: (1), 33-42.
- 187**§19** Hoffman, L.C., Wiklund, E. (2006). Game and venison: meat for the modern consumer. Meat Science, 74, 197-208. 188620
- Hoffman, L.C. (2013). Producing game meat to meet the International hygiene and food safety 188**621** requirements in the bush. American Meat Science Association, 66th Annual Reciprocal Meat 1882622 Conference (RMC) Auburn University, Auburn, ALA. 1883623
- Honkanen, P., Verplanken, B., Olsen, S.O. (2006). Ethical values and motives driving organic food 188 624 188525 choice. Journal of Consumer Behaviour, 5, 420-430.
- 188626 Hrubes, D., Ajzen, I., Daigle, J. (2001). Predicting hunting intentions and behavior: An application 1887627 of the Theory of Planned Behavior. Leisure Sciences, 23: (3), 165-178.
- ¹⁸⁸⁸028 Hutchinson, C.L., Mulley, R.C., Wiklund, E., Flesh, J.S. (2010). Consumer evaluation of venison ¹⁸⁸⁹629 sensory quality: Effects of sex, body condition score and carcase suspension method. Meat Science, 86: (2), 311-316.
- 1890 1891 1892 1892 1893 1893 1894 1894 1894 Italian National Institute of Statistics (ISTAT) (2018). ISTAT demographic balance January 2018. Rome.
 - Kaiser, H.F. (1974). An index of factorial simplicity. Psychometrika, 39, 31-36.
- 1895 Kellert, S.R., Berry, J.K. (1987). Attitudes, knowledge, and behaviors toward wildlife as affected by 189635 gender. Wildlife Societv Bulletin, 15, 363-371.
- Kendal, H.A., Lobao, L.M., Sharp, J.S. (2006). Public concern with animal well-being: Place, social 189\$36 structural location, and individual experience. Rural Sociology, 71: (3), 399-428. 189\$37
- Kubberød, E., Dingstag, Ø., Ueland, G.I, Risvik, E. (2006). The effect of animality on disgust 189**638** response at the prospect of meat preparation. An experimental approach from Norway. Food 190639 Quality and Preference, 17, 199-208. 190%640
- 1902641 Kwiecińska, K., Kosicka-Gębska, M., Gębski, J., Gutkowska, K. (2017). Prediction of the conditions 190342 for the consumption of game by polish consumers. Meat Science, 131, 28-33.
- 190443 Lusk, J.L., Tonsor, G.T. (2016). How meat demand elasticities vary with price, income, and product 190544 category. Applied Economic Perspectives and Policy, 38: (4), 673-711.
- 190645 Ljung, P.E., Riley, S.J., Heberlein, T. A. (2012). Eat Prey and Love: Game-meat consumption and 1907646 attitudes toward hunting. Wildlife Society Bulletin, 36: (4), 669-675.
 - Malhotra, N.K. (1999). Marketing research: an applied orientation. Upper saddle River, NJ: Prentice-Hall International.
- 1908 1908 1909 1910 1910 1911 1912 50 Mayfield, L.E., Bennet, R.M., Tranter, R.B., Wooldridge, M.J. (2007). Consumption of welfarefriendly food products in Great Britain, Italy and Sweden, and how it may be influenced by consumer attitudes to, and behavior towards, animal welfare attributes. International Journal of 191**§**51 Sociology of Food and Agriculture, 15: (3), 59-73. 191\$52
- Olsen, S.O. (2001). Consumer involvement in seafood as family meals in Norway: An application of 191553 191654 the expectancy-value approach. Appetite, 36: (2), 173-186.
- Olsen, S.O., Scholderer, J., Brunso, K., Verbeke, W. (2007). Exploring the relationship between 1917655 1918556 convenience and fish consumption: A cross-cultural study. Appetite, 49: (1), 84-91.
- 191%57 Pérez-Cueto, F.J.A., Pieniak, Z., Verbeke, V. (2011). Attitudinal determinants of fish consumption in 192058 Spain and Poland. Nutr. Hosp., 26, 1412-1419.
- 192%59 Peterson, R.A. (1994). A meta-analysis of Cronbach's coefficient alpha. Journal of Consumer 192260 Research, 21, 381-391.
- ¹⁹²³61 Pieniak, Z., Verbeke, W., Scholderer, J., Brunsø, K., Ottar Olsen, S. (2008). Impact of consumers' ¹⁹²⁴ 662 1925 663 1926 health beliefs, health involvement and risk perception on fish consumption: A study in five European countries. British Food Journal, 110: (9), 898-915.
- 1927

1928

- 1930
- ¹⁹³64 Pieniak, Z., Aertsens, J., Verbeke, W. (2010a). Subjective and objective knowledge as determinants of organic vegetables consumption. Food Quality and Preferences, 21, 581-588.
 - Pieniak, Z., Verbeke, W., Scholderer, J. (2010b). Health-related beliefs and consumer knowledge as determinants of fish consumption. Journal of human nutrition and dietetics, 23: (5), 480-488.

1932 1933 665 1933 666 1934 667 1935 668 1936 668 1936 669 Prestamburgo, S., Sgroi, F. (2018). Agro-food Markets' functional efficiency, products' quality and information's role. Calitatea – Access la Succes, 19: (164), 149.

- 193**6**70 Ramanzin, M., Amici, A., Casoli, C., Esposito, L., Lupi, P., Marsico, G., Mattiello, S., Olivieri, O., Ponzetta, M.P., Russo, C., Trabalza Marinucci, M.M. (2010). Meat from wild ungulates: ensuring 193671 194**6**72 quality and hygiene of an increasing resource. Italian Journal of Animal Science, 9: (61), 318-194673 331.
- Ripoll, G., Alberti, P., Panea, B. (2015). Consumer segmentation based on food-related lifestyles and 1942674 1943675 perception of chicken breast. International Journal of Poultry Science, 14: (5), 262-275.
- 194%676 Roininen, K.; Lähteenmäki, L; Tuorila, H. (1999). Quantification of consumer attitudes to health and 1945677 hedonic characteristics of foods. Appetite, 33, 71-88.01
- ¹⁹⁴678 Rule, D.C., Broughton, K.S., Shellito, S.M., Maiorano, G. (2002). Comparison of muscle fatty acid 1947679 profiles and cholesterol concentrations of bison, beef cattle, elk, and chicken. Journal of Animal ¹⁹⁴⁸80 Science, 80, 1202-1211.
 - Sepúlveda, W.S., Maza, M.T., Pardos, L. (2011). Ascpects of quality related to the consumption and production of lamb meat. Consumers versus producers. Meat Science, 87, 366-372.
- 1949 1949 1950 1950 1951 1952 1952 1952 1953 Swanepoel, M., Leslie, A.J., Hoffman, L.C. (2016). Comparative analyses of the chemical and sensory parameters and consumer preference of a semi-dried smoked meat product (cabanossi) 1954**8**5 produced with warthog (Phacochoerus africanus) and domestic pork meat. Meat Science, 114, 103-195586 113.
- Thogmartin, W. (2006). Why not consider the commercialization of deer harvests?. Bioscience, 56, 195687 957. 1957688
- 195\$89 Thulin, C.G., Malmsten, J., Ericsson, G. (2015). Opportunities and challenges with growing wildlife populations and zoonotic diseases in Sweden. European Journal of Wildlife Research, 61: (5), 195**%90** 649-266. 196691
- 196%692 Tomasevic, I., Novakovic, S., Solowiej, B., Zdolec, N., Skunca, D., Krocko, M., Nedomova, S., 196293 Kolaj, R., Aleksiev, G., Djekic, I. (2018). Consumers' perceptions, attitudes and perceived quality 196394 of game meat in ten European countries. Meat Science, 142, 5-13.
- 196495 Van Loo, E.J., Caputo, V., Nayga, R.M., Meullenet, J.F., Crandall, P.G., Ricke, S.C. (2010). Effect ¹⁹⁶⁵696 of organic poultry purchase frequency on consumer attitudes toward organic poultry. Journal of Food Science, 75: (7), S384-S397.
 - Van Loo, E.J., Hoang Diem, M.N., Pieniak, Z., Verbeke, W. (2013). Consumer attitudes, knowledge, and consumption of organic yogurt. Journal of Dairy Science, 96: (4), 2118-2129.
- 1966 97 1967 98 1967 98 1968 99 1969 700 1977 01 Van Wezemael, L., Verbeke, W., Kügler, J.O, de Barcellos, M.D., Grunert, K.G. (2010). European consumers and beef safety: Perceptions, expectations and uncertainty reduction strategies. Food Control, 21, 835-844. 1972702
- 197303 Vanhonacker, F., Verbeke, W., Van Pouche, E., Tuyttens, F.A.M. (2007). Segmentation based on consumers' perceived importance and attitude toward farm animal welfare. International Journal 197704 1973705 of Sociology of Food and Agriculture, 15: (3), 84-100.
- Vanhonacker, F., Van Loo, E.J., Gellynck, X., Verbeke, W. (2013). Flemish consumer attitudes 1970706 197707 towards more sustainable food choices. Appetite, 62, 7-16.
- 1978/08 Vecchio, R., Annunziata, A. (2012). Italian consumer awareness of layer hens' welfare standards: a 197909 cluster analysis. International Journal of Consumer Studies, 36, 647-655.
- 198910 Verbeke, W., Vackier, I. (2004). Profile and effects of consumer involvement in fresh meat. Meat 198711 Science, 67, 159-168.
- ¹⁹⁸712 Verbeke, W.; Viaene, J. (1999). Beliefs, attitude and behavior towards fresh meat consumption in ¹⁹⁸³713 1984 Belgium: empirical evidence from a consumer survey. Food Quality and Preference, 10, 437-445.
- 1985
- 1986
- 1987
- 1988

- Verbeke, W.; Pérez-Cueto, F.J.A.; de Barcellos, M.D.; Krystallis, A.; Grunert, K.G. (2010). European citizen and consumer attitudes and preferences regarding beef and pork. *Meat Science*, 84: (2), 284-292.
 Webb, F.C. (2003). Factors that affect the meat multiplication of the meat multiplication.
- Webb, E.C. (2003). Factors that affect the meat quality of domesticated animals and wildlife. *Game* and Hunt, 9: (7), 51.
- Willebrand, T. (2009). Promoting hunting tourism in north Sweden: opinions of local hunters. *European Journal of Wildlife Research*, 55, 209-216.
- **721** Wiklund, E., Farouk, M., Finstad, G. (2014). Venison: Meat from red deer (*Cervus elaphus*) and reindeer (*Rangifer tarandus tarandus*). *Animal Frontiers*, 4: (4), 55-61.
- Windle, J., Rolfe, J. (2011). Comparing responses from internet and paper-based collection methods
 in more complex stated preference environmental valuation surveys. *Economic Analysis and Policy*, 41: (1), 83-97.
- Wong, S.S., Aini, M.S. (2017). Factors influencing purchase intention of organic meat among consumers in Klang Valley, Malaysia. *International Food Research Journal*, 24: (2), 767-778.

1	Have you	consumed w	ild game me	at in the last v	voor?		
1.	Have you consumed who game meat in the last year :						
	0 Yes						
	O No						
2.	Please ind	licate wheth	er the followi	ng statements	s are true or f	false:	
-	"In Italy th	ne population	s of wild ungu	lates (red deer	r, wild boar, ro	e deer, cha	mois) are ra
	growing"	(True)					
-	"Hunters a	are not consid	lered producer	rs by the Italia	n law; therefo	ore they can	not sell the
	they hunt"	' (False)					
-	"Game me	eat (red deer,	wild boar, roe	e deer, chamoi	s) if compared	l with beef	meat has a
	protein co	ntent and hig	her fat and che	olesterol conte	ent" (False)		
-	"Consumi	ng game me	eat (red deer,	wild boar, r	oe deer, char	nois) is an	ethical c
	sustainable	e and with a v	very low envir	conmental imp	act" (True)		
-	"Currently	in Italy hunt	ing is 'control	led', in fact it	is subject to w	ell defined	rules conce
			1 1 22 (77				
	culling, sp	ecies, times a	ind places" (1	rue)			
3.	culling, sp For each	statements	please indicates indicates please indicates please indicates please indicates please indicates please pleas	rue) ate which ac	ljectives bett	er describ	e your fe
3.	culling, sp For each towards t	statements	please indication of HWG	<i>rue)</i> ate which ac M products.	ljectives bett	er describ	e your fe
3.	culling, sp For each towards t	statements he consumpt	nd places" (7 please indication ion of HWG	rue) ate which ac M products.	ljectives bett	er describ	e your fe Favoura
3. Unfa	culling, sp For each towards the avourable attisfied	statements he consumpt	nd places" (7 please indication of HWG)	rue) ate which ac M products.	ljectives bett O O	er describ	e your fe Favoura Satisfie
3. Unfa Unsa Disli	culling, sp For each towards th wourable atisfied	statements statements he consumpt	ind places" (7 please indication of HWG 0 0 0	rue) ate which ac M products.	ljectives bett O O O	er describ	e your fe Favoura Satisfie Like
3. Unfa Unsa Disli Nega	culling, sp For each towards t wourable atisfied ike ative	statements statements he consumpt	nd places" (7 please indication ion of HWG 0 0 0 0	rue) ate which ad M products.	ljectives bett O O O O	er describ	e your fe Favoura Satisfie Like Positive
3. Unfa Unsa Disli Nega Bad	culling, sp For each towards t avourable atisfied ike	statements statements he consumpt O O O O O O	nd places" (7 please indication ion of HWG 0 0 0 0 0	rue) ate which ad M products. 0 0 0 0	ljectives bett O O O O O O	er describ	e your fe Favoura Satisfie Like Positive Good
3. Unfa Unsa Disli Nega Bad Terri	culling, sp For each towards the towards t	statements statements he consumpt O O O O O O O O O O O	please indication of HWG	rue) ate which ad M products.	ljectives bett O O O O O O O O	er describ	e your fe Favour Satisfie Like Positive Good Delight
3. Unfa Unsa Disli Nega Bad Terri	culling, sp For each towards the avourable atisfied ike ative	statements statements he consumpt O O O O O O O O O	ind places" (7 please indication ion of HWG 0 0 0 0 0 0	rue) ate which ad M products. 0 0 0 0 0	ljectives bett O O O O O O O	er describ	e your fe Favour Satisfie Like Positive Good Delight
3. Unfa Unsa Disli Nega Bad Terri	culling, sp For each towards the avourable atisfied ike ative	statements statements he consumpt O O O O O O O O	please indication of HWG	rue) ate which ad M products.	ljectives bett O O O O O O O	er describ	e your fe Favour Satisfie Like Positive Good Delight

For each statements please indicate which adjective better describe your feelings when 3.

thinking about wild game meat safety.

Pessimistic	0	0	0	0	0	Optimistic
Unconfident	0	0	0	0	0	Confident
Unsatisfied	0	0	0	0	0	Satisfied
Uncomfortable	0	0	0	0	0	Comfortable
Suspicious	0	0	0	0	0	Trustful
Worrying	0	0	0	0	0	At ease

Would you agree or disagree with the following statements? (Those statements where 4.

not categorized and appeared in mixed order)

	Strongly disagree	Disagree	Neither disagree nor agree	Agree	Strongly agree
ANIMAL WELFARE (AW)					
It is important that the food I normally has been produced in a way that anim-	y eat mals O	0	0	0	0
It is important that the food I norm eat has been produced in a way	ally that O	0	0	0	0
In general humans have too l respect for the quality of life of anir	little O nals	О	0	0	О
Increased regulation of the treatment animals in farming is needed	nt of O	0	0	0	0
Animal agriculture raises ser ethical questions about the treatmen animals	ious nt of O	0	0	0	0
As long as animals do not suffer p humans should be able to use them any purpose	oain, O n for O	0	0	0	0
It is acceptable to use animals to consumer products such as so cosmetics and household cleaners	test aps, O	0	0	0	0

2166						
2167	Hunting animals for sport is an	0	0	0	0	0
2168	acceptable form of recreation	0	0	0	Ũ	0
2169	ATTITUDES TOWARDS HUNTING					
2170	(A II)					
2171					•	
2172	Hunting helps keep nature in balance	0	0	0	0	0
2174	Marthur an mall and a lark a					
2175	Most numers are wen-prepared when	0	0	0	0	0
2176	they go hunting					
2177	I see little wrong with harvesting	0	0	0	0	0
2178	animals for their meat as long as the	0	0	0	0	0
2179	animal is not endangered					
2180	Hunters are properly trained and follow	0	0	0	0	0
2181	hunting regulations					
2182	Hunting is an important rural tradition	0	0	0	0	0
2103	fruiting is an important futur tradition	0	0	U	Ũ	0
2185	I regard any kind of sport and	0	0	0	0	0
2186	recreational hunting as cruel to animals	0	0	0	0	0
2187	Hereten a ften innen as fate relat	0	0	•	0	0
2188	Hunters often ignore safety rules	0	0	0	0	0
2189	Unators often have an intels which	_	_	_	_	_
2190	Hunters often narm animals, which	0	0	0	0	0
2191	then dies a slow and painful death					
2192	I do not like people who hunt	0	0	0	0	0
2193						
2194	WILDLIFE-RELATED VALUES					
2195	(WV)					
2197	It is important to manage the	-	-	_	-	-
2198	populations of wildlife for benefit of	0	0	0	0	0
2199	humans					
2200	Leniov watching wildlife when I take a	0	0	0	0	0
2201	trin	0	0	0	0	0
2202	It is important to protect wildlife for				•	
2203	it is important to protect whether for	0	0	0	0	0
2204	tuture generations					
2205	Hunting and fishing are cruel and	0	0	0	0	0
2200	inhumane to the animals					
2208	I notice birds and wildlife around me	0	0	0	0	0
2209	every day					
2210	People should not cause pain and					
2211	suffering to wildlife, regardless of how	0	0	0	0	0
2212	much we may benefit					
2213	It is important that we learn all we can	•	•	0	0	•
2214		0	0	0	0	0
2215	about wildlife					
2210	Animals should have rights similar to	0	0	0	0	0
2217	the rights of humans					
221g752						
222753						
2227754						
2222						
2223						38

Appendix B:

356

Table B1. Results of the Factor analysis of the 37-items: rotated component matrix results

				Comp	oonent			
	Factor 1	Factor 2	Factor 3	Factor 4	Factor 5	Factor 6	Factor 7	Factor 8
GENERAL ATTITUDES (ATT)								
Favorable/Unfavorable (R)	0.690							
Unsatisfied/Satisfied	0.770							
Dislike/Like	0.838							
Negative/Positive	0.841							
Bad/Good	0.828							
Terrible/Delightful	0.837							
PERCEPTION OF SAFETY								
(SAF)								
Pessimistic/Ottimistic			0.773					
Unconfident/Confident			0.789					
Satisfied/Unsatisfied (R)			0.508					
Uncomfortable/ Comfortable			0.777					
Suspicious/Trustful			0.777					
Worrying/At ease			0.809					
			0.050					
ANIMAL WELFARE (AW) It is important that the food I								
normally eat has been produced in								
a way that animals have not		0.796						
experienced pain								
It is important that the food I								
normally eat has been produced in		0 701						
a way that animals' rights have		0.781						
been respected								
In general humans have too little								
respect for the quality of life of		0.705						
animals								
Increased regulation of the		0 774						
treatment of animals in farming is		0.774						
Animal agriculture raises serious								
ethical questions about the		0 729						
treatment of animals		0.727						
As long as animals suffer pain								
humans should not be able to use								0.80
them for any purpose (R)								
It is acceptable to use animals to								
test consumer products such as								0.64
soaps, cosmetics and household								0.04
cleaners (R)								
Hunting animals for sport is an								0.47
acceptable form of recreation (R)								0.1/.
ATTITUDES TOWARDS								
HUNTING (AH)								

2284				
2285	Hunting helps keep nature in			
2286	halance	0.606		
2287	Most hunters are well-prepared			
2288	when they go hunting	0.784		
2289	I see little wrong with harvesting			
2290	animals for their meat as long as	0 548		
2291	the animal is not endangered	0.010		
2292	Hunters are properly trained and			
2293	follow hunting regulations	0.758		
2294	Hunting is an important rural			
2295	tradition	0.663		
2296	I regard any kind of sport of			
2297	recreational hunting as cruel to		0.722	
2298	animals (R)			
2299	Hunters often ignore safety rules		0.910	
2300	(R)		0.819	
2301	Hunters often harm animals,			
2302	which then dies a slow and painful		0.761	
2303	death (R)			
2304	I do not like people who hunt (R)		0.502	
2305	WILDLIFE-RELATED VALUES			
2306	(WV)			
2307	It is important to manage the			
2308	populations of wildlife for benefit		-0.516	
2309	of humans (R)			
2310	I enjoy watching wildlife when I		0.714	
2311	take a trip		0.714	
2312	It is important to protect wildlife		0.605	
2313	for future generations		0.005	
2314	Hunting and fishing are cruel and			0 708
2315	inhumane to the animals			0.708
2316	I notice birds and wildlife around		0.614	
2317	me every day		0.014	
2318	People should not cause pain and			
2319	suffering to wildlife, regardless of	0.588		
2320	how much we may benefit			
2321	It is important that we learn all we		0 609	
2322	can about wildlife		0.007	
2323	Animals should have rights			0.622
2324	similar to the rights of humans			

Table B2. PCA components internal reliability consistency

Seelo	Cronbach's α-values				
State	Original scale	PCA Subscale 1	PCA Subscale 2	Reduced scale	
GENERAL ATTITUDES (ATT)	0.946	-	-		
PERCEPTION OF SAFETY (SAF)	0.919	-	-		
ANIMAL WELFARE (AW)	0.696	0.863 (Factor 2)	0.550 (Factor 8)		
ATTITUDES TOWARD HUNTING (AH)	0.866	0.835 (Factor 4)	0.816 (Factor 5)		
WILDLIFE RELATED VALUES (VW)	0.830	0.515 (Factor 6)	0.686 (Factor 7)	0.654*	









